Author's response to reviews

Title: Genomewide Association Study for Onset Age in Parkinson Disease

Authors:

Jeanne C Latourelle (jlatoure@bu.edu)
Nathan Pankratz (npankrat@iupui.edu)
Alexandra Dumitriu (adumitri@bu.edu)
Jemma B Wilk (jwilk@bu.edu)
Stefano Goldwurm (goldwurm@parkinson.it)
Gianni Pezzoli (Pezzoli@parkinson.it)
Claudio B Mariani (mariani@parkinson.it)
Anita L Destefano (adestef@bu.edu)
Cheryl Halter (chalter@iupui.edu)
James F Gusella (gusella@helix.mgh.harvard.edu)
William C Nichols (Bill.Nichols@cchmc.org)
Richard H Myers (rmyers@bu.edu)
Tatiana Foroud (tforoud@iupui.edu)

Version: 3 Date: 31 August 2009

Author's response to reviews: see over
August 27, 2009

Dear Editors,

We are pleased to hear that you have accepted in principle our manuscript ‘Genomewide Association Study for Onset Age in Parkinson Disease’.

As requested, we have removed the following reference on page 18.


This reference has been replaced with two more comprehensive references that more directly support the discussion point being made.


We have also checked the formatting to conform to the BMC Journal’s standards. Please let us know if any additional revisions are required.

Sincerely,

[Signature]

Fatima [Surname]
Tatiana Foroud, Ph.D.
(317) 278-1291

Richard Myers, Ph.D.
(617) 638-5376